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Beam	
  status	
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Spectrometer	
  status	
  
•  Beam	
  vacuum	
  pipe	
  –	
  stable	
  
•  Cerenkov	
  counter	
  

–  Cerenkov	
  mirror	
  replaced	
  
–  2	
  neutral	
  density	
  filters	
  replaced	
  

•  Target	
  –	
  stable	
  (LH2,	
  LD2,	
  C,	
  No	
  
target)	
  
–  microSD	
  card	
  on	
  flammable	
  gas	
  

sensor	
  replaced	
  aTer	
  “trouble”	
  
signal	
  

•  Hodoscopes	
  	
  
–  H1R5,	
  H2R5	
  PMT	
  bases	
  replaced	
  

•  Chambers	
  
–  D0,	
  D1,	
  D2,	
  D3p,	
  D3m	
  –	
  stable	
  	
  

•  Prop	
  tubes	
  –	
  P42V	
  trip	
  recovered	
  
•  Trigger	
  –	
  stable	
  
•  Online	
  monitoring	
  –	
  stable	
  
•  DAQ	
  –	
  stable	
  

–  ROC	
  19	
  had	
  a	
  bad	
  TDC	
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CompuFng	
  status	
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CONTINUE	
  GOOD	
  DATA	
  TAKING!	
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